Author's response to reviews

Title: Preferred HIV testing services and programme characteristics among clients of a rapid HIV testing programme.

Authors:

Juan Hoyos (jhoys@isciii.es)
María J Belza (mbelza@isciii.es)
Sonia Fernández-Balbuena (soniafb@isciii.es)
María E Rosales (milenanstatkus@yahoo.com)
J Pulido (jpulido@isciii.es)
L de la fuente (lfuente@isciii.es)
Madrid HIV rapid testing group (rbksan@isciii.es)

Version: 4 Date: 14 June 2013

Author's response to reviews: see over
Dear Sir/Madam:

Please find enclosed a file containing the comments of the changes made to our article “Preferred HIV testing services and programme characteristics among clients of a rapid HIV testing programme.”

The manuscript has undergone revision following the suggestions made by reviewer nº1 (reviewer nº2 had no further comment). You will find his comments, followed by our replies. Where appropriate, page and line are detailed. These changes are highlighted in green. The manuscript has also been revised to enhance written English as suggested by the editor.

We hope that, with these modifications, the article will meet your requirements for publication.

**Reviewer 1: Ulrich Marcus**

Reviewer’s report: I am mostly satisfied with the revisions made. There is one minor point which is still unsatisfying. While in Table 1 "sex work" has been changed into "ever exchanged money for sex", this has not been done in Table 3 and in the Results section (page 6). Also, since "ever exchanged money for sex" may cover paying as well as being paid for sex, changing this to "ever been paid for sex" may be the better option.

As suggested by the reviewer, “ever exchanged sex for money” has been changed to “ever been paid for sex”, both in the manuscript (Results section 1st page, line 6) and in table 1 and 3.